Author’s response to reviews

Title: Individual, Social Environmental, and Physical Environmental Correlates with Physical Activity among Canadians: a Cross-sectional Study

Authors:

Sai Yi Pan (sai_yi_pan@phac-aspc.gc.ca)
Christine Cameron (ccameron@cflri.ca)
Marie DesMeules (marie_desmeules@phac-aspc.gc.ca)
Howard Morrison (Howard_Morrison@phac-aspc.gc.ca)
Cora Lynn Craig (ccraig@cflri.ca)
Xiaohong Jiang (Xiaohong_Jiang@phac-aspc.gc.ca)

Version: 5 Date: 28 October 2008

Author’s response to reviews:

Typographical formatting for the footnotes of table 2, 3, and 4 has been made.